\begin{thebibliography}{10}

\bibitem{microreboot:osdi04}
G.~Candea, S.~Kawamoto, Y.~Fujiki, G.~Friedman, and A.~Fox.
\newblock Microreboot \&\#8212; a technique for cheap recovery.
\newblock In {\em Proceedings of the 6th Conference on Symposium on Opearting
  Systems Design \& Implementation - Volume 6}, OSDI'04, pages 3--3, Berkeley,
  CA, USA, 2004. USENIX Association.

\bibitem{nice:nsdi12}
M.~Canini, D.~Venzano, P.~Pere\v{s}\'{\i}ni, D.~Kosti\'{c}, and J.~Rexford.
\newblock A nice way to test openflow applications.
\newblock In {\em Proceedings of the 9th USENIX Conference on Networked Systems
  Design and Implementation}, NSDI'12, pages 10--10, Berkeley, CA, USA, 2012.
  USENIX Association.

\bibitem{devoflow:sigcomm11}
A.~R. Curtis, J.~C. Mogul, J.~Tourrilhes, P.~Yalagandula, P.~Sharma, and
  S.~Banerjee.
\newblock Devoflow: Scaling flow management for high-performance networks.
\newblock In {\em Proceedings of the ACM SIGCOMM 2011 Conference}, SIGCOMM '11,
  pages 254--265, New York, NY, USA, 2011. ACM.

\bibitem{ofpostcard:nsdi14}
N.~Handigol, B.~Heller, V.~Jeyakumar, D.~Mazi{\`e}res, and N.~McKeown.
\newblock I know what your packet did last hop: Using packet histories to
  troubleshoot networks.
\newblock In {\em Proceedings of the 11th USENIX Symposium on Networked Systems
  Design and Implementation (NSDI 14)}, pages 71--85, Seattle, WA, 2014.
  USENIX.

\bibitem{sdnlayerdebugging:hotsdn13}
B.~Heller, C.~Scott, N.~McKeown, S.~Shenker, A.~Wundsam, H.~Zeng, S.~Whitlock,
  V.~Jeyakumar, N.~Handigol, J.~McCauley, K.~Zarifis, and P.~Kazemian.
\newblock Leveraging sdn layering to systematically troubleshoot networks.
\newblock In {\em Proceedings of the Second ACM SIGCOMM Workshop on Hot Topics
  in Software Defined Networking}, HotSDN '13, pages 37--42, New York, NY, USA,
  2013. ACM.

\bibitem{b4:sigcomm13}
S.~Jain, A.~Kumar, S.~Mandal, J.~Ong, L.~Poutievski, A.~Singh, S.~Venkata,
  J.~Wanderer, J.~Zhou, M.~Zhu, J.~Zolla, U.~H\"{o}lzle, S.~Stuart, and
  A.~Vahdat.
\newblock B4: Experience with a globally-deployed software defined wan.
\newblock In {\em Proceedings of the ACM SIGCOMM 2013 Conference on SIGCOMM},
  SIGCOMM '13, pages 3--14, New York, NY, USA, 2013. ACM.

\bibitem{fgft:asplos13}
A.~Kadav, M.~J. Renzelmann, and M.~M. Swift.
\newblock Fine-grained fault tolerance using device checkpoints.
\newblock In {\em Proceedings of the Eighteenth International Conference on
  Architectural Support for Programming Languages and Operating Systems},
  ASPLOS '13, pages 473--484, New York, NY, USA, 2013. ACM.

\bibitem{veriflow:hotsdn12}
A.~Khurshid, W.~Zhou, M.~Caesar, and P.~B. Godfrey.
\newblock Veriflow: Verifying network-wide invariants in real time.
\newblock In {\em Proceedings of the First Workshop on Hot Topics in Software
  Defined Networks}, HotSDN '12, pages 49--54, New York, NY, USA, 2012. ACM.

\bibitem{onix:osdi10}
T.~Koponen, M.~Casado, N.~Gude, J.~Stribling, L.~Poutievski, M.~Zhu,
  R.~Ramanathan, Y.~Iwata, H.~Inoue, T.~Hama, and S.~Shenker.
\newblock Onix: A distributed control platform for large-scale production
  networks.
\newblock OSDI'10.

\bibitem{shadowsdn:hotsdn13}
M.~Ku\'{z}niar, P.~Pere\v{s}\'{\i}ni, N.~Vasi\'{c}, M.~Canini, and
  D.~Kosti\'{c}.
\newblock Automatic failure recovery for software-defined networks.
\newblock HotSDN '13, pages 159--160, New York, NY, USA, 2013. ACM.

\bibitem{pease-1980-paxos}
M.~Pease, R.~Shostak, and L.~Lamport.
\newblock {Reaching Agreement in the Presence of Faults}.
\newblock {\em Journal of the ACM}, 27(2):228--234, 1980.

\bibitem{bigswitch}
{Big Switch Networks, Inc.}
\newblock \url{http://www.bigswitch.com}.

\bibitem{bigtap}
{Big Tap Monitoring Fabric}.
\newblock \url{http://bigswitch.com/products/big-tap-monitoring-fabric}.

\bibitem{criu}
{CRIU}.
\newblock \url{http://criu.org/Main_Page}.

\bibitem{flowscale}
{FlowScale}.
\newblock \url{http://www.openflowhub.org/display/FlowScale/FlowScale+Home}.

\bibitem{flowscale_bugs}
{FlowScale Issues}.
\newblock \url{ http://bugs.openflowhub.org/}.

\bibitem{opendaylight}
{OpenDaylight: A linux foundation collaborative project.}
\newblock \url{http://www.opendaylight.com}.

\bibitem{floodlight}
{Project Floodlight.}
\newblock \url{http://www.projectfloodlight.org/}.

\bibitem{floodlight_greatestecosystem}
{Project Floodlight Grows to the World's Largest SDN Ecosystem}.
\newblock \url{http://www.bigswitch.com/press-releases/
  2013/03/26/project-floodlight-grows-to-the-worlds-
  largest-sdn-ecosystem-global-users}.

\bibitem{fattire:hotsdn13}
M.~Reitblatt, M.~Canini, A.~Guha, and N.~Foster.
\newblock Fattire: Declarative fault tolerance for software-defined networks.
\newblock HotSDN '13, New York, NY, USA. ACM.

\bibitem{failureobliviouscomputing:osdi04}
M.~Rinard, C.~Cadar, D.~Dumitran, D.~M. Roy, T.~Leu, and W.~S. Beebee, Jr.
\newblock Enhancing server availability and security through failure-oblivious
  computing.
\newblock In {\em Proceedings of the 6th Conference on Symposium on Opearting
  Systems Design \& Implementation - Volume 6}, OSDI'04, pages 21--21,
  Berkeley, CA, USA, 2004. USENIX Association.

\bibitem{routeflow:hotsdn12}
C.~E. Rothenberg, M.~R. Nascimento, M.~R. Salvador, C.~N.~A. Corr\^{e}a,
  S.~Cunha~de Lucena, and R.~Raszuk.
\newblock Revisiting routing control platforms with the eyes and muscles of
  software-defined networking.
\newblock In {\em Proceedings of the First Workshop on Hot Topics in Software
  Defined Networks}, HotSDN '12, pages 13--18, New York, NY, USA, 2012. ACM.

\bibitem{shadowdrivers:tcs06}
M.~M. Swift, M.~Annamalai, B.~N. Bershad, and H.~M. Levy.
\newblock Recovering device drivers.
\newblock {\em ACM Trans. Comput. Syst.}, 24(4):333--360, Nov. 2006.

\bibitem{nox:hotnets09}
A.~Tavakoli, M.~Casado, T.~Koponen, and S.~Shenker.
\newblock Applying nox to the datacenter.
\newblock In L.~Subramanian, W.~E. Leland, and R.~Mahajan, editors, {\em
  HotNets}. ACM SIGCOMM, 2009.

\bibitem{rulebricks:hotsdn13}
D.~Williams and H.~Jamjoom.
\newblock Cementing high availability in openflow with rulebricks.
\newblock In {\em Proceedings of the Second ACM SIGCOMM Workshop on Hot Topics
  in Software Defined Networking}, HotSDN '13, pages 139--144, New York, NY,
  USA, 2013. ACM.

\end{thebibliography}

\section{Introduction}
%SDN embodies a novel paradigm that is poised to revolutionize the networking industry, with research predicting a market size of \$3.5 billion~\cite{}. Despite its burgeoning ecosystem and proven benefits, many companies remain hesitant in adopting it. Two of the foremost issues that hamper to SDN adoptions are reliability and fault tolerance~\cite{}. These issues are only bound to become even more important as societies grows increasingly dependent  on cloud computing and online social networks.

Despite Software Defined Network's, SDN, proven benefits, many companies remain hesitant in adopting it. Two of the foremost issues that hamper to SDN adoptions are reliability and fault tolerance~\cite{}. \asim{I remember reading security is the biggest problem}. In a traditional SDN deployment, there are four main failure scenarios: controller server failures (hardware failures), controller crashes (bugs in the controller code), network device failure (switch, application server or link failure), and application crashes (bugs in the application code). While much work has gone into debugging the SDN ecosystem~\cite{sdnlayerdebugging:hotsdn13, ofpostcard:nsdi14, veriflow:hotsdn12, nice:nsdi12}, little work has gone into hardening SDN again these failures.

Most recent work focuses on overcoming hardware failures, namely controller server failures~\cite{onix:osdi10, b4:sigcomm13} and network device failures~\cite{shadowsdn:hotsdn13,fattire:hotsdn13}. Unfortunately, software failures remain un-addressed.  Furthermore, traditional operating system and application-server fault-tolerant techniques like reboot or replay~\cite{microreboot:osdi04,shadowdrivers:tcs06} can not simply be applied to the SDN control plane for three reasons.  

First, both the network and the applications both contain state, rebooting~\cite{microreboot:osdi04} the application will eliminate this state leading to inconsistency issues.  Second, unlike with in an OS, where-in a web-server can be rebooted without affecting other applications on the same OS, in an \SDN environment, the \SDN app is intermingled with the controller, so a reboot affects all other apps.  Third, replaying events~\cite{shadowdrivers:tcs06} to recover this state implicitly assumes that the bugs are non-deterministic and thus wouldn't be encountered during replay. However, we argue that given the event driven nature of \SDN applications, bugs will most likely be deterministic. In fact a brief survey of controller and application forums confirms this.
\asim{this is a very strong claim: all events, no threading. are you sure a single application does not have multiple threads in a controller library?}
% asim : shadowdrivers:tcs06 removed: I REMOVED THE SHADOW DRIVER REFERENCE, shadow drivers infact, do replay and driver code is also mashed with OS. you are better off sticking with applications in this section.


In this work, we focus on application crashes because the controller code represents a common layer that is highly re-used and thus has a lower probability of bug, whereas the application code is more likely to be provided by third party entities with limited testing.  A trend that is expected to become even more prevalent given the recent success of project \floodlight~\cite{floodlight_greatestecosystem} and the development OpenDayLight project~\cite{opendaylight}.   Furthermore, as stated earlier the lack of a clean separation between applications and controllers, introduces a fate-sharing relationship where-in an application crash induces a controller crash.  


The principle position of this paper is that availability is the upmost concern -- second only to perhaps security. To this end, we propose a re-architecture of the controller architecture that eliminates fate-sharing between applications through isolation and that relaxes the correctness of the polices implemented by an application with a {\it best-effort model}. We argue for a model where-in applications may not process all network events, 
to maintain controller availibility.

%We argue that strictly enforcing correctness requires that the application process all network event; however, if these events themselves trigger failures then the application is unable to satisfy both correctness and availability. In certain situations, we believe that the availability is a better choice than blindly enforcing correctness. \asim{there goes the determinism?}

The challenge lies in determining when to compromise correctness, how much to comprise, and how to ensure safety while compromising correctness. Where-in safely is defined as: at most one copy of a message is sent from the application to the network that the application is aware of all messages sent to the network. Thus during an application rollback, the network must also rollback. To overcome these challenges, we leverage properties of \SDN applications, namely that they are event driven.  This allows us to isolate and detect crash-triggering events and in turn determine when to compromise.  In addition given the nature of network events, we can allow the operator to choose between various levels of comprise. For example, if a link down event triggers an application crash. The operator can choose compromise 100\% by ignoring the event, or 0\% compromise by crashing the application, or something in between by transforming the port down event into a switch down event.  Finally, for safety, we implement a network roll-back mechanism which leverages the insight that most \SDN events are invertible. \asim{is this true for all events, then say all SDN events are invertible} For example, installing a rule is invertible by deleting the rule.

%
%Given the strong desire to maintain networks with five 9s of availability, we argue that such compromises are acceptable. 
%
%Furthermore, we show that given the appropriate architecture, application isolation and application fault tolerance complement each.

% Furthermore, a redesign of the controller architecture enables a large array of features currently absent.
%
%\name builds on \appvisor and aims to target both Bohrbugs bugs (deterministic) and Heisenbugs bugs (non-deterministic).
%
To this end, we propose \name a system that embodies the described principles.  \name provides isolation layer between applications and introduces a fault tolerance layer that detects crash triggering events and overcomes them through event transformations.  In \name applications run unmodified, thus shielding application programmers from having to understand the functionality and recovery techniques provided by \name.
%
\name provides a policy language for operators allowing them to specify the set of applications and network events whose accuracy and efficacy they are willing to compromise on.
%
This enables operators to trade-off between control plane available and correctness of network policy.
%
Thus allowing the network to persevere in the face of application crashes and provide slightly degraded functionality until a fix or patch can be deployed.
%
As a proof-of-concept, we re-architected the \floodlight controller to support \name. Using this prototype, we demonstrate that \name imposes minimal overheads and requires no application modifications.


\breadcrumbs{The rest of this paper is structured as follows: in section~\ref{}, XXX.}
\comments{
%{\it Migrating Applications Between Controllers:} Processor migration is a well understood technique in the O.S community. By treating individual controller apps as processes, we can leverage these migration techniques to move apps between distributed controllers. This would be particularly useful as a strategy to combat over utilized servers.

\begin{table}	
	\vspace{-0.05in}
	\begin{center}
\begin{tabular}{cc}
%$$\begin{figure}
		%\noindent \hrulefill 
		\epsfig{file=figs/FPGFlowChart.eps,height=2in} 
%\end{figure}

&

%\begin{figure}
		%\noindent \hrulefill 
		\epsfig{file=figs/ShadowApp.eps,height=2in} 
%\end{figure}
\end{tabular}
\vspace{-0.1in}
\caption{(a) Flow chart encoding fault tolerant module functionality. (b) \name Architecture}
	
\label{f:overload}
	\end{center}
\vspace{-0.35in}
\end{table}
}
